Author’s response to reviews

Title: Developing nephrology services in low income countries: A case of Tanzania

Authors:

Francis Furia (fredrick.francis78@gmail.com)
Jacqueline Shoo (shoojacqueline@gmail.com)
Paschal Ruggajo (prugajo@yahoo.com)
Kajiru Kilonzo (mtundumliasi@googlemail.com)
Gopal Basu (B.Gopal@alfred.org.au)
Karen Yeates (yeatesk@queensu.ca)
Santosh Varughese (santosh@cmcvellore.ac.in)
Einar Svarstad (Einar.Svarstad@uib.no)
Onesmo Kisanga (oakisanga@yahoo.co.uk)

Version: 2 Date: 24 Sep 2019

Author’s response to reviews:

Minor Comments from Reviewer 2 have been addressed as recommended.

Technical editorial comments have been addressed as explained below:

- List of abbreviations is provided
- Statement about ethical approval is provided
- Statement about availability of data sets is provided
- List